Author's response to reviews

Title: Association between subjective memory complaints and health care utilisation: a three-year follow up

Authors:

Frans Boch Waldorff (fbw@gpmed.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
Gunhild Waldemar (gunhild.waldemar@rh.regionh.dk)

Version: 6 Date: 23 September 2009

Author's response to reviews:

Dear editor,

We have noticed that the editorial production team has changed the order of the tables. However, we would like to keep Table 4 as the last table, because it seems more logic to us to present the multivariate analysis after the baseline tables. Thus, we deleted the presentation of the present Table 4 in the statistical section.

The author Volkert Siersma has changed his email and we have subsequently changed his email in the paper.

Best

Frans Waldorff